Author's response to reviews

Title: The effects of a web-based decision aid on the intention to diagnostic self-testing for cholesterol and diabetes: A randomized controlled trial

Authors:

Janaica EJ Grispen (janaica@hotmail.com)
Martine HP Ickenroth (martineickenroth@hotmail.com)
Nanne K De Vries (nanne.devries@maastrichtuniversity.nl)
Geert-Jan Dinant (geertjan.dinant@maastrichtuniversity.nl)
Gaby Ronda (gaby.ronda@maastrichtuniversity.nl)
Trudy Van der Weijden (trudy.vanderweijden@maastrichtuniversity.nl)

Version: 2
Date: 16 January 2014

Author's response to reviews:

To the editors of BMC Public Health

Dear editor,

We have made the required changes to our manuscript.

1.) We included the email address of all the authors on the Title page.
2.) We included a Competing interests section after the Conclusions.
3.) We included an Authors' Contributions section after the Competing Interests section.
4.) We included an Acknowledgements section after the Authors' Contributions section.

Kind regards on behalf of all authors,

Gaby Ronda

Maastricht University
CAPHRI School for Public Health and Primary Care
Department of General Practice
P.O. Box 616
6200 MD Maastricht, The Netherlands
Email: gaby.ronda@maastrichtuniversity.nl
T + 31 43 3882318